Author's response to reviews

Title: Precocious Puberty in a 4 year Old Egyptian Girl Caused By Mixed Germ Cell-Sex Cord-Stromal- Tumour associated with yolk sac tumour of the Ovary. a case report

Authors:

kotb A Metwalley (kotb72@yahoo.com)
Dalia A Elsers (daliaelsers@yahoo.com)
Hekma S Fargalley (hekma73@hotmail.com)
Hanaa A Mohamed (hae50@hotmail.com)
Mohamed A Othman (dr.othman@ymail.com)

Version: 2 Date: 15 April 2012

Author's response to reviews:

all the requested corrections by reviewers were done